Author's response to reviews

Title: Acute hospital-based services used by adults during the last year of life in New South Wales, Australia: a population-based retrospective cohort study

Authors:

David E Goldsbury (davidg@nswcc.org.au)
Dianne L O'Connell (dianneo@nswcc.org.au)
Afaf Girgis (afaf.girgis@unsw.edu.au)
Anne Wilkinson (anne.wilkinson@ecu.edu.au)
Jane L Phillips (jane.phillips@uts.edu.au)
Patricia M Davidson (pdavidson@jhu.edu)
Jane M Ingham (jane.ingham@svha.org.au)

Version: 6 Date: 19 November 2015

Author's response to reviews:

Thank you for accepting our revised manuscript. In accordance with your instructions, we have added the specific roles of the authors in the Authors' Contributions section, as shown below.

Page 27, from line 641: “DLO and JMI conceived and designed the study. DLO and DEG acquired the data. DEG analysed the data with input from DLO and JMI. DEG, DLO and JMI drafted the manuscript. All authors (DEG, DLO, AG, AW, JLP, PMD and JMI) were involved in revising the manuscript and all read and approved the final manuscript.”